\documentclass[a4paper,10pt,oneside]{article}

\marginparwidth=15pt
\oddsidemargin=10pt
\textwidth=440pt
\textheight=600pt

% PACKAGES
\usepackage[latin1]{inputenc}


% RAVI's FONT.
%\usepackage{mathpazo}
%\usepackage{euler}


\newcommand{\marginalfootnote}[1]{
   \footnote{#1}
   \marginpar{\hfill {\sf\thefootnote}}}
\newcommand{\edit}[1]{\marginalfootnote{#1}}


\usepackage[T1]{fontenc}
\usepackage[UKenglish]{babel}
\usepackage{amsfonts}
\usepackage{amssymb}
\usepackage{amsthm}
\usepackage{graphicx}
\usepackage{fancyhdr}
\usepackage{color}
\usepackage{makeidx}
\usepackage{amsmath}
\usepackage[all,cmtip]{xy}
\usepackage{blindtext}
\usepackage{parskip}
\usepackage{hyperref}
\usepackage[notlof,notlot]{tocbibind}
\usepackage{mathrsfs}

\addtocounter{section}{1}

%\setlength{\parindent}{0pt}
%\setlength{\headheight}{15.2pt}
%\renewcommand{\sectionmark}[1]{\markboth{#1}{}}
%\lhead{{\thepage}}
%\chead{}
%\rhead{{\textup{\leftmark}}}
%\lfoot{}
%\cfoot{}
%\rfoot{}

\makeindex

\definecolor{red}{rgb}{1,0,0}
\definecolor{lblue}{rgb}{0.3,0.4,1}
\definecolor{orange}{rgb}{1,0.6,0.2}
\definecolor{green}{rgb}{0.1,0.7,0.1}

% MEANINGS
\newtheoremstyle{thm}% name of the style to be used
  {3pt}% measure of space to leave above the theorem. E.g.: 3pt
  {3pt}% measure of space to leave below the theorem. E.g.: 3pt
  {\em}% name of font to use in the body of the theorem
  {0pt}% measure of space to indent
  {\bfseries}% name of head font
  {}% punctuation between head and body
  {5pt}% space after theorem head
  {}% Manually specify head
\newtheoremstyle{rem}% name of the style to be used
  {3pt}% measure of space to leave above the theorem. E.g.: 3pt
  {3pt}% measure of space to leave below the theorem. E.g.: 3pt
  {}% name of font to use in the body of the theorem
  {0pt}% measure of space to indent
  {\bfseries}% name of head font
  {}% punctuation between head and body
  {5pt}% space after theorem head
  {}% Manually specify head


% MACROS
\theoremstyle{thm}
%\newtheorem*{them}{Theorem}
\newtheorem{thm}{Theorem}[subsection]
\newtheorem{prop}[thm]{Proposition}%[subsection]
%\newtheorem*{propn}{Proposition}
\newtheorem{cor}[thm]{Corollary}%[subsection]
\newtheorem{lemma}[thm]{Lemma}%[subsection]
\theoremstyle{rem}
\newtheorem{eg}[thm]{Example}%[subsection]
\newtheorem{defin}[thm]{Definition}%[subsection]
\newtheorem{rem}[thm]{Remark}%[subsection]


% ADDITIONAL SYMBOLS
\newcommand\ab{{\rm ab}}
\newcommand\ad{{\rm ad}}
\newcommand\aff{{\rm aff}}
\newcommand\alg{{\rm alg}}
\newcommand\ant{{\rm ant}}
\newcommand\codim{{\rm codim}}
\newcommand\diag{{\rm diag}}
\newcommand\et{{\rm et}}
\newcommand\gr{{\rm gr}}
\newcommand\gp{{\rm gp}}
\newcommand\coker{{\rm coker}}
\newcommand\id{{\rm id}}
\newcommand\lin{{\rm lin}}
\renewcommand\log{{\rm log}}
\newcommand\op{{\rm op}}
\newcommand\ord{{\rm ord}}
\newcommand\red{{\rm red}}
\newcommand\reg{{\rm reg}}
\newcommand\rk{{\rm rk}}

\newcommand\Aff{{\rm Aff}}
\newcommand\Aut{{\rm Aut}}
\newcommand\Bl{{\rm Bl}}
\newcommand\End{{\rm End}}
\newcommand\GL{{\rm GL}}
\newcommand\Ga{{\mathbb G}_a}
\newcommand\Gm{{\mathbb G}_m}
\newcommand\Gal{{\rm Gal}}
\newcommand\Gr{{\rm Gr}}

\newcommand\Hilb{{\rm Hilb}}
\newcommand\Quot{{\rm Quot}}


\newcommand\fX{{\mathfrak{X}}}
\newcommand\fY{{\mathfrak{Y}}}
\newcommand\ccF{{\rm \mathcal{F}}}
\newcommand\ccD{{\rm \mathcal{D}}}
\newcommand\ccO{{\rm \mathcal{O}}}

\newcommand\Spf{\mathrm{Spf}}
\newcommand\cSpec{\underline{\mathrm{Spec}}}
\newcommand\disc{{\mathrm{disc}}}
\newcommand\Frob{{\mathrm{Frob}}}
\newcommand\rR{{\mathrm{R}}}
\newcommand\Tate{{\mathrm{Tate}}}
\newcommand\fG{{\mathfrak{G}}}
\newcommand\fH{{\mathfrak{H}}}

\newcommand\Hom{{\rm Hom}}
\newcommand\Ind{{\rm Ind}}
\newcommand\Irr{{\rm Irr}}
\newcommand\Lie{{\rm Lie}}
\newcommand\M{{\rm M}}
\newcommand\Mor{{\rm Mor}}
\newcommand\Pic{{\rm Pic}}
\newcommand\Proj{{\rm Proj}}
\newcommand\PGL{{\rm PGL}}
\newcommand\PSL{{\rm PSL}}
\newcommand\SL{{\rm SL}}
\newcommand\Spec{{\rm Spec}}
\newcommand\Spa{{\rm Spa}}
\newcommand\Sp{{\rm Sp}}
\newcommand\Supp{{\rm Supp}}
\newcommand\Sym{{\rm Sym}}
\newcommand\Univ{{\rm Univ}}
\newcommand\Ass{{\rm Ass}}
\newcommand\Frac{{\rm Frac}}
\newcommand\Tor{{\rm Tor}}
\newcommand\cTor{{\mathcal Tor}}
\newcommand\cHom{{\underline{\rm Hom}}}
\newcommand\RSchemes{{\bfseries aff} $R$-{\bfseries Sch}}
\newcommand\Schemes{{\mathbf{Sch}}}
\newcommand\Top{{\mathbf{Top}}}
\newcommand\Cov{{\mathrm{Cov}}}
\newcommand\Sets{\mathbf{Set}}
\newcommand\Func{\mathbf{Func}}
\newcommand\Pref{{\bfseries Presh}_X}
\newcommand\Sheaf{{\bfseries She}_X)}
\newcommand\Shv{\mathbf{Shv}}
\newcommand\PShv{\mathbf{PShv}}
\newcommand\Open{{\mathbf{Op}}_X}
\newcommand\Ab{{\mathbf{Ab}}}
\newcommand\Grps{\mathbf{Grp}}
\newcommand\RMod{\mathbf{Mod}_R}
\newcommand\RAlg{\mathbf{Alg}_R}
\newcommand\Ob{\textup{Ob}}
\newcommand\bA{{\mathbb A}}
\newcommand\bC{{\mathbf C}}
\newcommand\bF{{\mathbf F}}
\newcommand\bG{{\mathbb G}}
\newcommand\bI{{\mathbf I}}
\newcommand\bN{{\mathbf N}}
\newcommand\bP{{\mathbb P}}
\newcommand\bQ{{\mathbf Q}}
\newcommand\bR{{\mathbf R}}
\newcommand\bS{{\mathbb S}}
\newcommand\bV{{\mathbf V}}
\newcommand\bX{{\mathbf X}}
\newcommand\bZ{{\mathbf Z}}

\newcommand\bmu{{\mathbb \mu}}

\newcommand\cA{{\mathscr A}}
\newcommand\cB{{\mathscr B}}
\newcommand\cC{{\mathscr C}}
\newcommand\cD{{\mathscr D}}
\newcommand\cE{{\mathscr E}}
\newcommand\cF{{\mathscr F}}
\newcommand\cG{{\mathscr G}}
\newcommand\cH{{\mathscr H}}
\newcommand\cI{{\mathscr I}}
\newcommand\cJ{{\mathscr J}}
\newcommand\cL{{\mathscr L}}
\newcommand\cM{{\mathscr M}}
\newcommand\cN{{\mathscr N}}
\newcommand\cO{{\mathscr O}}
\newcommand\cQ{{\mathscr Q}}
\newcommand\cR{{\mathscr R}}
\newcommand\cS{{\mathscr S}}
\newcommand\cT{{\mathscr T}}
\newcommand\cV{{\mathscr V}}
\newcommand\cW{{\mathscr W}}
\newcommand\cX{{\mathscr X}}
\newcommand\cY{{\mathscr Y}}
\newcommand\cZ{{\mathscr Z}}
\newcommand\cProj{{\mathcal Proj}}
\newcommand\fg{{\mathfrak g}}
\newcommand\fh{{\mathfrak h}}
\newcommand\ft{{\mathfrak t}}
\newcommand\fD{{\mathfrak D}}
\newcommand\fS{{\mathfrak S}}
\newcommand\fm{{\mathfrak m}}
\newcommand\fn{{\mathfrak n}}
\newcommand\fp{{\mathfrak p}}
\newcommand\fq{{\mathfrak q}}
\newcommand\ccU{{\mathfrak U}}
\newcommand\cU{{\mathscr U}}
\newcommand\bJ{{\mathbf J}}
\newcommand\bH{{\mathbf H}}
\newcommand\fA{{\mathfrak A}}
\newcommand\fB{{\mathfrak B}}
\newcommand\fC{{\mathfrak C}}
\newcommand\SR{{\fS(R_{\mathrm{fpqc}})}} % descent 1


\newcommand\tg{\tilde{g}}
\renewcommand\th{\tilde{h}}
\newcommand\tlambda{\tilde{\lambda}}
\newcommand\tp{\tilde{p}}
\newcommand\tq{\tilde{q}}

\newcommand\tG{\tilde{G}}
\newcommand\tM{\tilde{M}}
\newcommand\tV{\tilde{V}}
\newcommand\tX{\tilde{X}}
\newcommand\tY{\tilde{Y}}
\newcommand\tZ{\tilde{Z}}

\newcommand\tcZ{\tilde{\mathcal Z}}

\newcommand\ualpha{\underline{\alpha}}
\newcommand\ulambda{\underline{\lambda}}

\newcommand\BT{\mathbf{BT}}
\newcommand\fotimes{{\widehat{\otimes}}}

\newcommand\Inf{{\mathrm{Inf}}}
\newcommand\uomega{{\underline{\omega}}}
\newcommand\Ext{{\mathrm{Ext}}}
\newcommand\rH{{\rm H}}
\newcommand\rV{{\rm V}}
\newcommand\rT{{\rm T}}
\newcommand\Ann{{\rm Ann}}
\newcommand\Spv{{\rm Spv}}
\newcommand\RZ{{\rm RZ}}


%\renewcommand{\sectionmark}[1]{\markright{\thesection\ #1}}
%\fancyhf{}  % delete current setting for header and footer
%\fancyhead[LE,RO]{\thepage}
%\fancyhead[LO]{\bfseries\rightmark}
%\fancyhead[RE]{\bfseries\leftmark}
%\renewcommand{\headrulewidth}{0.5pt}
%\renewcommand{\footrulewidth}{0pt}
%\addtolength{\headheight}{0.5pt} % make space for the rule
%\fancypagestyle{plain}{%
%   \fancyhead{} % get rid of headers on plain pages
 %  \renewcommand{\headrulewidth}{0pt} % and the line
%}

\begin{document}
\begin{thebibliography}{dieci}
%\bibitem[Bosch]{Bosch} S S. Bosch, {\em Lectures on Formal and Rigid Geometry}. Springer Lecture Notes in Mathematics, vol. 2105.
%\bibitem[Con]{Con} BB. Conrad, ``Several approached to non-archimedean geometry'', Arizona Winter School lecture notes
%available at: \texttt{http://math.stanford.edu/$\sim$conrad/papers/aws.pdf}.
\bibitem[H]{hartshorne} RR.\,Hartshorne, {\em Algebraic Geometry}, GTM 52, Springer-Verlag, 1977.
\bibitem[Mat]{Mat} HH. Matsumura, {\em Commutative Ring Theory}.
Vol. 8 of Cambridge Studies in Advanced Mathematics (1989).
%\bibitem[MM]{MM} SS. MacLane, I. Moerdijk,
%{\em Sheaves in Geometry and Logic: a first introduction to topos theory}. Springer Science $\&$ Business Media, 1992.
%\bibitem[vdP]{vdP} C M. van der Put, {\em Cohomology
%on affinoid spaces.} Compositio Math. 45 (1982), 165 - 198.
%\bibitem[vdPS]{vdPS} P M. van der Put, P. Schneider, {\em Points of rigid analytic varieties.} J. reine angew. Math. 434 (1993), 127 - 157.
\bibitem[T]{temkin} MM.\,Temkin, {\em Relative Riemann--Zariski spaces}, Israel Journal of Math {\bf 85} (2011), 1--42.
\bibitem[Zar1]{Zar1} OO. Zariski, {\em The reduction of the singularities of an algebraic surface}. Ann. of Math. (2) 40 (3): 639-689 (1939).
\bibitem[Zar2]{Zar2} OO. Zariski, {\em Reduction of the singularities of algebraic three dimensional varieties}. Ann. of Math. (2) 45 (3) (1944).
\end{thebibliography}

\normalsize
\section{Lecture 2: Spaces of valuations}
\subsection{Introduction}

Let $A$ be a commutative ring. Recall that
$$\Spec(A)=\{x: A\to K\}/\sim$$
that is, $\Spec(A)$ is the set of equivalence classes of homomorphisms $x:A \to K$ to fields $K$, where two such are considered {\em equivalent} as soon as they can be compatibly embedded into third field (giving the same homomorphism on $A$).
Indeed, we can shrink $K$ so that we can assume $K$ is generated by the image of $A$ under $x$. Thus, we have
$$0\to\fp\to A\to x(A)\to 0$$
where $\fp$ is the kernel of $x$, and is a prime ideal, as $x(A)$ is a domain whose fraction field is 
the target of $x$. We therefore recover our usual notion of $\Spec$ of a ring as the set of prime ideals (with a suitable topology on this set). 
Note that if $A$ is finitely generated over a field $k$ and we consider only $x$ making $K := {\rm{Frac}}(x(A))$ of finite degree over $k$
then we recover the more classical MaxSpec construction in such cases.

Similarly, if we fix a non-archimedean field $k$, and a $k$-affinoid algebra $A$, we can view the Berkovich spectrum of $A$ as
$$M(A)=\{ x: A\to K\ \textup{continuous}\}/\sim;$$
that is, the set of continuous homomorphisms $A\to K$ into non-archimedean
field extensions of $k$ (with compatible absolute value), taken up to a suitable equivalence.
In contrast with the classical rigid-analytic use of MaxSpec,
here we most definitely do {\em not} require $K$ to be $k$-finite. 

Though it may seem ungeometric at first, with adic spaces 
we are going to work with {\em higher-rank} valuations on a field, valued in some totally ordered abelian group. As we shall discuss, this additional freedom will allow the theory of certain {\em spaces of valuations}, which we begin to introduce in this lecture. \\

This lecture is organized as follows. We give an overview of the theory of valuation rings, the main reference for this being 
\cite[\S10]{Mat}. Then, we review the classical notion of Riemann--Zariski space for fields,
and then reach the valuation spectrum of a commutative ring $A$ as a generalization.
Although our ultimate interest is in the case of certain topological rings and valuations
satisfying a continuity condition, 
the quasi-compactness properties of such refined spaces of continuous valuations we shall introduce 
later will follow from those of the valuation spectrum $\Spv(A)$ of general commutative rings $A$.
Hence, we first study $\Spv$ in this lecture and the next one before moving on to the case of
topological rings closer to those of interest in non-archimedean geometry.
%\footnote{Please email \texttt{amasullo@stanford.edu} about any typo or mistake contained in these notes.}




\subsection{Valuation rings}\label{valsec}
\begin{defin} \label{def: valuation}
For a field $K$, a {\em valuation} on $K$ is a map
$$v: K\to \Gamma\cup\{0\}$$
where $\Gamma$ is a totally ordered abelian group with the additional rules:
$$0<\gamma,\ \ \forall\ \gamma\in\Gamma$$
$$0 \cdot 0 = 0, \,\,\,\gamma\cdot 0=0\ \ \forall\gamma\in\Gamma,$$ which satisfies the following properties:
\begin{enumerate}
\renewcommand{\labelenumi}{\textup{(\arabic{enumi})}}
\item $v(0)=0$, $v(1)=1$.
\item $v(xy)=v(x)v(y)$, for all $x,y\in K$.
\item $v(x+y)\leq\max(v(x), v(y))$, for all $x,y\in K$.
\end{enumerate}
Note that $v(x) \ne 0$ for all $x \in K^{\times}$ since $v(1) = 1$.
\end{defin}


\begin{eg} The following assignment
$$v(x):=\left\{\begin{array}{cc}
0 & x=0\\
1 & x\neq 0
\end{array}\right.
$$
is called the {\em trivial} valuation on $K$.
\end{eg}


\begin{rem} Often $v$ is denoted
with the symbol $|\cdot |$; our definition is actually that of an {\em absolute value} on $K$, but since we will
be using higher-rank cases it is traditional not to call such things ``absolute values''.
\end{rem}

\begin{rem}\label{arch}
We say a valuation $v$ on $K$ has {\em rank $1$} if it is non-trivial and $\Gamma\subset\bR^{\times}_{>0}$ as ordered
abelian groups, and it has {\em higher rank} if it is nontrivial and not of rank 1.
Sometimes valuations that are either trivial or rank 1 (i.e., $\Gamma \subset \bR^{\times}_{>}$ as ordered
abelian groups) are called {\em archimedean} (a potentially confusing terminology!) because 
an ordered abelian group $\Gamma$ is a subgroup of $\bR^{\times}_{>0}$ if and only if for 
every $\gamma < 1$ in $\Gamma$ the powers $\{\gamma^n\}_{n > 0}$ are cofinal for the order
structure: for any $\gamma'$ there exists $n > 0$ such that $\gamma^n < \gamma'$; see \cite[Thm.\,10.6]{Mat}.
(This is the multiplicatively reciprocated formulation of the classical ``archimedean property'' in the axioms for the real line.)
\end{rem}

\begin{rem} It is harmless to shrink $\Gamma$ to $v(K^{\times})$.
This is technically convenient to avoid  getting side-tracked into the notion of ``equivalence'' for valuations.
This will become more relevant as we move on to valuations on rings.
\end{rem}

We recall that, given a field $K$ and a valuation $v$ on $K$, we are naturally given a subring of $K$:
$$R_v:=\{x\in K\mid v(x)\leq 1\}$$
the {\em valuation ring} of $v$ in $K$. The units $R_v^{\times}$ of $R_v$ are easily seen to be
$$\{x\in K\mid v(x)=1\}.$$
As a direct consequence of this, it follows that $R_v-R_v^{\times}$ is an ideal of $R_v$, which we shall denote $\fm_v$, and therefore $R_v$ is a local ring with unique maximal ideal $\fm_v$.

We can give the notion of valuation ring of a field $K$ without mentioning the valuation on $K$,
as follows. 
From the above discussion, we notice that a property of $R_v$ is that for all $x\in K^{\times}$, either $x\in R_v$ or $x^{-1}\in R_v$. It also follows that $\Frac(R_v)=K$.

\begin{defin} \label{def: valuation bis} A {\em valuation ring} is a domain $R$, with fraction field $K$, such that for all $x$ in $K^{\times}$, either $x\in R$ or $x^{-1}\in R$.
\end{defin}

We can rephrase the Definition \ref{def: valuation bis} in terms of divisibility of non-zero
elements of $R$, that is: for all $a,b\in R - \{0\}$, either $a$ divides $b$ or $b$ divides $a$. This latter reformulation has the immediate consequence that as soon as an ideal of a valuation ring $R$, with fraction field $K$, is finitely generated, then it is principal.

To relate Definition \ref{def: valuation bis} back to \ref{def: valuation}, 
we need to explain how a valuation ring as in Definition \ref{def: valuation bis} 
comes implicitly with an associated valuation and a value group. Observe that if $R$ is a valuation ring with fraction field $K$, then 
$$\Gamma_R:=K^{\times}/R^{\times}$$
is a totally ordered abelian group where, for all $a,b\in K^{\times}$, we say
$$a\ \textup{mod}\ R^{\times} \leq b\ \textup{mod}\ R^{\times}$$
if and only if $x:=a/b$ is in $R$. We are naturally given a map
$$v:K\to\Gamma_R\cup\{0\}$$
by letting $v(a)$ be $a\ \textup{mod}\ R^{\times}$  if $a \ne 0$ or $0$ if $a = 0$.

\begin{rem} There may well exist more than one valuation $v$ on $K$ with valuation ring $R$ and value group isomorphic to $\Gamma$. 
For example, let $K=\bQ$, and $v_{p,\varepsilon}$ be the $p$-adic valuation on $\bQ$, that is, given an $0<\varepsilon<1$,
$$v_{p,\varepsilon}(a/b):=\varepsilon^{-(\textup{ord}_p(a)-\textup{ord}_p(b))}$$
for nonzero $a, b \in \mathbf{Z}$.
For all such $\varepsilon$, the value group is always {\em isomorphic} to $\bZ$. The valuations $v_{p,\varepsilon}$ are 
 distinct $\bR^{\times}_{>0}$-valued {\em functions} on $\bQ$, but they yield the {\em same} valuation ring $\bZ_{(p)}$.  By regarding $v$
 as having target $v(K^{\times})$, the data of the map $v$ and totally ordered abelian group
 $\Gamma$ are absorbed into the data of the valuation ring itself.
 Hence, it is really the valuation ring that is the fundamental structure of interest.
\end{rem}


\paragraph{A recipe to produce higher rank valuations}\ \\
Let $K$ be a field equipped with a valuation, and call $R$ its valuation ring, and $\fm$ its unique maximal ideal. Let us assume the residue field $k:=R/\fm$ is equipped again with a valuation. We call
$$\overline{R}'\subset k$$
its corresponding valuation ring. We define the subset of $R$
$$R':=\{x\in R\mid x\ \textup{mod}\ \fm \in \overline{R}'\}.$$
Clearly $R'$ is a subring of $R$. We have $\fm\subset R'\subset R$, and $x'\in R'$ lies in ${R'}^{\times}$ if and only if its
reduction in $k$ is in ${\overline{R}'}^{\times}$.

Assume $\fm\neq (0)$. Let $a$ and $b$ be in $R-\{0\}$, and pick $c\in\fm-\{0\}$. Since
$$\frac{a}{b}=\frac{ca}{cb},$$
we have $\Frac(R')=K$. We observe that in the case $\fm=(0)$, this is true as well. 

\begin{prop}\label{prop:valbar} $R'$ is a valuation ring of $K$.
\end{prop}

We call $R'$ the {\em composite} of the valuation rings $R$ and $\overline{R}'$.

\begin{proof} Choose $x\in K^{\times}$. If $x\in R'$, then we are done, so let us assume $x$ is not in $R'$. We must show $x^{-1}\in R'$. We first let $x$ be such that $x\notin R$. Then, since $R$ is a valuation ring, $x^{-1}\in R$, and this latter is not a unit. It follows $x^{-1}\in R'$.\\
Now, if $x\in R - R'$, then $\overline{x}:=x\ \textup{mod}\ \fm$ is not in $\overline{R}'$. Since $\overline{R}'$ is a valuation ring of $k$, it follows $\overline{x}^{-1}\in\overline{R}'$. Since $\fm$ is contained in $R'$, so $x$ is a unit in $R$, 
we have that $x^{-1}\in R$ and this reduces to an element of $\overline{R}'$, modulo $\fm$. Thus, $x^{-1}\in R'$ as desired.
\end{proof}

In the Appendix, we introduce the notion of {\em convex} subgroup of a totally ordered abelian group,
discuss a total order structure modulo such a subgroup,
and explain the relationship between the orderings on $\Gamma_R$, $\Gamma_{\overline{R}'}$, and 
$\Gamma_{R'}$: 
\begin{thm}\label{thm:valgp} The value group $\Gamma_{\overline{R}'}$
is naturally a subgroup of $\Gamma_{R'}$, and as such is a convex subgroup
for which there is a natural isomorphism of ordered abelian groups $\Gamma_{R'}/\Gamma_{\overline{R}'} \simeq \Gamma_R$.
\end{thm}

The procedure described in the above discussion is, in fact, exhaustive in the following sense (see \cite[Thm.\,10.1]{Mat}):
\begin{thm}\label{thm:local}
 Let $R$ be a valuation ring with fraction field $K$. The process described above produces all valuation subrings $R'$ of $R$ with fraction field $K$.  Moreover, the maximal ideal $\mathfrak{m}$ of $R$ is a prime ideal of $R'$ and $R = R'_{\mathfrak{m}}$.
\end{thm}

\medskip

Let $R$ be a valuation ring with fraction field $K$. Let  $\Gamma:=\Gamma_R=K^{\times}/R^{\times}$ be the value group of $R$, and $v$ the valuation on $K$ given by reduction modulo $R^{\times}$. Some properties of $R$ can be read off from $\Gamma$. For example, 
since $v(R-\{0\})$ is contained in $$\Gamma_{\leq 1}:=\{\gamma\in\Gamma\mid \gamma\leq 1\}$$
we see that ideals of $R$ give
rise to submonoids of $\Gamma$ contained in $\Gamma_{\leq 1}$ under the mapping
$$I\mapsto v(I\setminus\{0\})=:M_I.$$
Note that if $I=(0)$ then $M_I$ is empty.\\

We leave to the reader as an exercise to prove that this is an inclusion-preserving bijection from the set of ideals
of $R$ onto the set of submonoids $M \subset \Gamma_{\le 1}$ such that $m \in M, \gamma \le m \Rightarrow \gamma \in M$.
(Informally, such $M$ behave like intervals in $\mathbf{R}$ that are unbounded to the left.)
The condition that $I$ is a finitely generated nonzero ideal of $R$ (or equivalently,
principal and nonzero) is exactly the condition that $M$
has a ``right endpoint'': that is, $M=\Gamma_{\leq\gamma_0}$ for some $\gamma_0\in\Gamma_{\leq 1}$.\\

An immediate consequence of the inclusion-preserving correspondence $I\mapsto M_I$ between ideals of $R$ and 
submonoids of $\Gamma$ that are ``unbounded to the left''
is that the set of ideals of $R$ is totally ordered set with respect to inclusion (as this property is readily verified to hold
for the corresponding collection of monoids: if $M$ and $M'$ are two such monoids and $M \not\subset M'$
then for $m \in M$ not in $M'$ we cannot have $m \le m'$ for an $m' \in M'$, so by the total ordering
necessarily $m' \le m$ and hence $m' \in M$ for all $m' \in M'$, which is to say $M' \subset M$).

\paragraph{A few general facts about valuation rings}\ \\
The following are proved in \cite[\S10]{Mat}:
\begin{enumerate}
\renewcommand{\labelenumi}{\textup{(\arabic{enumi})}}
\item Any valuation ring $R$ is integrally closed (proof exactly as in the case of discrete valuation rings).
\item If we have a local inclusion of valuation rings $R\subset R'$, with fraction fields $K$ and $K'$ respectively, then we have an induced inclusion of the corresponding value groups:
$$\Gamma_R\subset\Gamma_{R'},$$
as totally ordered abelian groups (i.e., if $\gamma, \delta \in \Gamma_R$ then $\gamma \le \delta$ in $\Gamma_R$ if and only if
$\gamma \le \delta$ in $\Gamma_{R'}$).
\item Valuation rings with fraction field $K$ are exactly the local domains contained in $K$ maximal with respect to local domination.
\item Let $A$ be a domain with fraction field $K$. Then the normalization of $A$ is the intersection of all the valuation rings of $K$ containing $A$, that is:
$$\widetilde{A}=\bigcap_{R\supset A}R,$$
the intersection running over all valuation rings of $K$ containing $A$.
\end{enumerate}



\subsection{Riemann-Zariski spaces for fields}
\begin{defin} \label{def: RZ} Let $A$ be an integral domain, and $K$ field containing it (eg., $K=\Frac(A)$). We say a vauation ring $R$ of $K$ with ${\rm{Frac}}(R) = K$
is {\em centered} in $A$ when $R$ contains $A$. In which case, the prime ideal $\fm_R\cap A$ is called the {\em center} of $R$ in $A$. The set of valuation rings of $K$ centered in $A$ is called the {\em Riemann-Zariski space} of
$K$ with respect to $A$, and will be denoted by $\RZ(K, A).$
\end{defin}
Throughout the next lectures we will be studying more intricate spaces of valuations, but $\RZ(K,A)$ will be the prototype, and indeed 
Riemann-Zariski spaces for fields will play a useful technical role even in later generalizations. 
The Riemann-Zariski space ${\rm{RZ}}(K,A)$ has a topology as follows:  for $x_1,...,x_n\in K$ we define
$$U(x_1,...,x_n):=\RZ(K, A[x_1,...,x_n]).$$
More explicitly, since $\RZ(K, A[x_1,...,x_n])$ is, by definition, the set of valuation rings of $K$ centered at $A[x_1,...,x_n]$, 
$U(x_1,...,x_n)$ is nothing but 
$$\{v\ \textup{valuation on}\ K\mid A \subset R_v, v(x_i)\leq 1,\ i=1,...,n\}.$$
Trivially, we have
$$U(x_1,...,x_s)\cap U(y_1,...,y_t)=U(x_1,...,x_s,y_1,...,y_y),$$
which ensures that the family
$$\cF:=\{U(x_1,...,x_n)\mid n\geq 0, x_i\in K\}$$
indeed generates a topology on $\RZ(K, A)$. In analogy with the case of $\Spec(A)$, this topology is called {\em Zariski topology} on 
$\RZ(K,A)$.  The special case $A = \bZ$ gives the set of all valuations on $K$ (as any valuation on $\bZ$ takes values $\le 1$
since $\bZ$ is additively generated by 1), and we write $\RZ(K)$ to denote $\RZ(K,\bZ)$.


\begin{rem} In the case $K = {\rm{Frac}}(A)$, the basic open sets $U(x_1,...,x_n)$ have a simpler form. Via finite intersections, we have that the topology on $\RZ(K)$ is actually generated by sets of the form
$$U(x):=\{v\ \textup{valuation on}\ K\mid A \subset R_v, v(x)\leq 1\}$$
with $x \ne 0$.
Since $x\in K$, 
there exist $a$ and $b$ in $A$ with $b \ne 0$ such that $x=a/b$, and so the topology on $\RZ(A)$ is generated by basic open sets of the form
$$U(a,b):=\{v\ \textup{valuation on}\ K\mid A \subset R_v, v(a)\leq v(b)\}.$$
The condition $b \ne 0$ can equally well be written as the condition $v(b) \ne 0$.
By writing instead
$$U(a,b) = \{v\ \textup{valuation on}\ K \mid A \subset R_v, v(a) \leq v(b) \ne 0\},$$
we see that the right side also makes sense when $b = 0$ but $U(a, 0) = \emptyset$.
This may seem like a silly remark, but it will be of (slightly) more interest when we pass
to valuations on rings (in which case there will be lots of non-zero non-units $b$).
\end{rem}

\begin{eg} Let us take $A=K$. Then, according to Definition \ref{def: RZ}, $\RZ(K,K)$ is simply a one-point space. 
\end{eg}

\begin{eg} Let $K=k(\!(y)\!)(\!(x)\!) = {\rm{Frac}}(k(\!(y)\!)[\![x]\!])$ for a field $k$, 
and $A=k[\![x]\!]$. Then $\RZ(K,A)$ is the set of those valuations $v$ on $K$ 
trivial on $k$ such that $v(x)\leq 1$, by property $(3)$ of Definition \ref{def: valuation}. One such example is the 
evident ``$x$-adic'' discrete valuation $v$. Its valuation ring $R$ is $k(\!(y)\!)[\![x]\!]$. Let us put in practice the procedure described 
in connection with Proposition \ref{prop:valbar}. The maximal ideal $\fm\subset R$ is $xR$, so that, if we equip $R/\fm\simeq k(\!(y)\!)$ with the 
evident $y$-adic valuation  (trivial on $k$)
then we get $\Gamma_R = x^{\bZ} \simeq\bZ$, $\Gamma_{\overline{R}'}= y^{\bZ} \simeq\bZ$, and the exact sequence 
of abelian groups 
$$1 \rightarrow \Gamma_{\overline{R}'} \rightarrow \Gamma_{R'} \rightarrow \Gamma_R \rightarrow 1$$
naturally splits to give $$\Gamma_{R'}\simeq \Gamma_R\times\Gamma_{\overline{R}'}\simeq \bZ^2$$
in which (exercise!) $\bZ^2$ is totally ordered with the lexicographical ordering 
that $(n,m)< (n',m')$ if and only if either $n<n'$ or $n=n'$ and $m<m'$. This value group violates
the ``archimedean property'' as in Remark \ref{arch}, so $R'$ is a higher-rank valuation ring.
\end{eg}

We have the following:
\begin{thm} $\RZ(K,A)$ is quasi-compact with respect to the Zariski topology.
\end{thm}
This is achieved via a Zorn's Lemma argument to establish the ``finite intersection property''
criterion. See \cite[Thm.\,10.5]{Mat} for the details.

\medskip

\begin{rem} Zariski deeply studied ${\rm{RZ}}(K,k)$ for a finitely generated
extension of fields $K/k$, especially with $k$ algebraically closed of characteristic 0.  He gave a classification of the 
valuation rings of $K$ containing $k$ (see \cite[Ch.\,II, Exer.\,4.12(b)]{hartshorne} for the 2-dimensional case). 
This enabled him to prove the resolution of singularities of algebraic varieties in dimensions 2 and 3 
over such $k$; see \cite{Zar1}, and \cite{Zar2}. 
Hironaka's general proof of resolution of singularities in characteristic 0 was obtained with other methods, with no use of valuation theory.

In general, if $K/k$ is a finitely generated extension of fields then the collection $\{X_i\}$ of projective integral $k$-schemes
with function field $K$ is an inverse system and the valuation criterion for properness
provides a compatible system of maps $\RZ(K,k) \rightarrow X_i$ under which the preimage of
a quasi-compact open subspace is a quasi-compact open subspace
(the preimage of an affine open ${\rm{Spec}}(B) \subset X_i$ is $\RZ(K,B)$).
Zariski proved that the resulting continuous map $\RZ(K,k) \rightarrow \varprojlim X_i$ is a homeomorphism.
Recent work of Temkin (see \cite{temkin}) 
has revived the role of valuation theory and (generalizations of) Riemann-Zariski spaces in the study
of resolution problems in the scheme-theoretic framework; e.g., see \cite[Cor.\,3.4.7]{temkin} for a vast generalization of Zariski's
inverse limit description of $\RZ(K,k)$, applicable to any relatively affine morphism between arbitrary qcqs schemes
and inspired by Huber's work on adic spaces.
\end{rem}

\subsection{$\Spv(A)$ as a Riemann-Zariski space}

\paragraph{Valuations on commutative rings}\ \\
We are ready to generalize the notion of valuation on a field $K$ to a commutative ring $A$.
\begin{defin} \label{def: valuationcommring} Let $A$ be a commutative ring. A 
{\em valuation} on $A$ is the datum of a map $v:A\to\Gamma\cup\{0\}$, where $\Gamma$ is a totally ordered abelian group, with the additional compatibility rule between $\Gamma$ and $0$ as in Definition \ref{def: valuation}, satisfying the following properties:
\begin{enumerate}
\renewcommand{\labelenumi}{\textup{(\arabic{enumi})}}
\item $v(0)=0, v(1)=1$.
\item $v(xy)=v(x)v(y),$ for all $x,y\in A$.
\item $v(x+y)\leq\max(v(x), v(y))$ for all $x,y\in A$.
\end{enumerate}
If $A = 0$ then there is no such $v$, and if $A \ne 0$ then it admits a homomorphism to a field,
on which the trivial valuation pulls back to one on $A$.  Of course, if $A$ is not a field
then $A - \{0\}$ is larger than $A^{\times}$, so it is no longer the case that necessarily $v(x) \ne 0$ when $x \ne 0$.
We define
$$\textup{supp}(v):=\{x\in A : v(x)=0\}=v^{-1}(0)$$
the {\em support} of $v$, which is easily seen to be a prime ideal, due to Definition \ref{def: valuationcommring}.
\end{defin}
Letting $\fp_v$ be the support of $v$, naturally $v$ induces a well-defined valuation $\widetilde{v}$ on 
the residue field $\kappa(\fp_v)$ at $\fp_v$. Let $\Gamma_v:=\widetilde{v}(\kappa(\fp_v)^{\times})$. Write $R_v$ for the valuation ring of $\widetilde{v}$ in $\kappa(\fp_v)$, so that $$\Gamma_v=\kappa(\fp_v)^{\times}/R_v^{\times}.$$ It is not at all true that, in general, $v(A)\subset\Gamma_{\leq 1}\cup\{0\}$, as shown by the following example:
\begin{eg}
Let $A=K$ a field. Then, we recover our old notion of valuation, as in Definition \ref{def: valuation}. If $v$ is nontrivial, then $K=A\nsubseteq R_v$.  Every field algebraic over a finite field admits only the trivial valuation (as all nonzero elements have
finite order yet a totally ordered abelian group is torsion-free), but any other field does admit (many!)
non-trivial valuations.  Indeed, any such field contains
either $\bQ$ or $\mathbf{F}_p(x)$ for some prime $p$, and 
valuations on a subfield always extend to any extension field (perhaps at the cost of a huge value group) due to the
characterization of valuation rings as maximality with respect to local domination in a field.
\end{eg}
\begin{defin} \label{def: equiv valuations} Two valuations $v$ and $v'$ on $A$
are {\em equivalent} if the following equivalent conditions are satisfied:
\begin{enumerate}
\renewcommand{\labelenumi}{\textup{(\arabic{enumi})}}
\item There is an isomorphism $\Gamma_{v}\simeq\Gamma_{v'}$ $($necessarily unique$)$ such that the following diagram commutes
\begin{displaymath}
\xymatrix{  A\ar[d]\ar[dr] & \\
		\Gamma_{v}\cup\{0\}\ar[r]_{\sim} & \Gamma_{v'}\cup\{0\}}
\end{displaymath}
\item $\fp_v=\fp_{v'}$, and $R_v=R_{v'}$ as valuation rings of the residue field at this common prime ideal of $A$;
\item For all $x,y\in A$, $v(x)\geq v(y)$ if and only if $v'(x)\geq v'(y)$.
\end{enumerate}
\end{defin}

We view condition (2) as the most efficient means of encoding a valuation on a ring $A$ without the hassle of ``equivalence'': 
we simply pick a prime ideal $\mathfrak{p}$ of $A$ and choose a valuation subring $R$ of $\kappa(\mathfrak{p})$ with 
${\rm{Frac}}(R) = \kappa(\mathfrak{p})$.

\begin{defin} \label{def: Spv} Let $A$ be a commutative ring. The {\em valuation spectrum} of $A$, denoted $\textup{Spv}(A)$,
is the set of equivalence classes of valuations on $A$ (more concretely: the set of pairs $(\mathfrak{p}, R)$
where $\mathfrak{p}$ is a prime ideal of $A$ and $R \subset \kappa(\mathfrak{p})$ is a valuation subring with full fraction field).
The set 
$\textup{Spv}(A)$ is made into a topological space by letting the topology being generated by the sets 
$$U\left(\frac{f}{g}\right):=\{v\in \Spv(A) \mid v(f)\leq v(g)\neq 0\},$$
for $f,g\in A$.  (The set
$U(f/g)$ is empty if $g = 0$, but for functoriality purposes as we vary $A$ it is good to allow this silly case in the definition.)
\end{defin}

In other words a base for the topology is given by finite intersections of sets $U(f/g)$. Informally, 
$U(f/g)$ should be viewed as the set of valuations $v$ for which $v(f/g) \le 1$, even though the fraction $f/g$ doesn't generally make sense
in $A$.  This notation is borrowed from work with rational domains in rigid-analytic geometry.

\begin{rem}
It may initially seem hard to visualize the topology on $\textup{Spv}(A)$. For example, if
$v, w \in \Spv(A)$ then what does it means that $v \in \overline{\{w\}}$ (i.e., $v$ is a ``specialization'' of $w$)?
We will see later how to probe the topology with some
concrete specialization constructions (generalizing Proposition \ref{prop:valbar} and Riemann-Zariski spaces of fields). 
To build a structure sheaf we will need to enrich our assumptions on $A$ and restrict the class of valuations  on $A$ that we consider.
\end{rem}

A good way to organize our study of the topology of $\Spv(A)$ is to think about the natural map 
$$\xymatrix{ \Spv(A) \ar[d]^-{\varphi} \\ {\rm{Spec}}(A)}$$
sending a valuation $v$ on $A$ to its support $\fp_v$. 
As a warm-up, suppose ${\rm{Spec}}(A)$ is open in a proper scheme $X$ over a field $k$ (thereby making $A$ a $k$-algebra) 
and $v$ is trivial 
on $k$, so the valuation ring $R_v \subset \kappa(\mathfrak{p}_v)$ is a $k$-subalgebra.
The valuative criterion for properness ensures that the $k$-morphism ${\rm{Spec}}(\kappa(\mathfrak{p}_v)) \rightarrow X$
extends (uniquely) to a $k$-morphism ${\rm{Spec}}(R_v) \rightarrow X$, and this lands inside the open subscheme 
${\rm{Spec}}(A)$ if and only if
the image of $A$ inside $\kappa(\mathfrak{p}_v)$ lands inside $R_v$; generally it doesn't!  That is, the closed point of
${\rm{Spec}}(R_v)$ typically will land outside ${\rm{Spec}}(A)$ in $X$.  Geometrically speaking, $\varphi(v)$ is the
image in $X$ of the {\em generic point} of ${\rm{Spec}}(R_v)$, {\em not} the image of the closed point
of ${\rm{Spec}}(R_v)$.

\begin{prop} The map $\varphi$ is continuous with respect to the respective topologies on $\Spv(A)$ and $\Spec(A)$.
\end{prop}
\begin{proof} Indeed, consider a basic open set $D(b)$ in $\Spec(A)$. Then $$\varphi^{-1}(D(b))=\{v\in\Spv(A)\mid v(b)\neq 0\}=\{v\in\Spv(A)\mid v(0)\leq v(b)\neq 0\}$$
which is $U\left(\frac{0}{b}\right)$, and is open.
\end{proof}

The fiber of $\varphi$ over any point $\fp\in\Spec(A)$ is topologically
the set $\RZ(\kappa(\fp)) = \RZ(\kappa(\mathfrak{p}),\bZ)$ equipped
with its Zariski topology as defined earlier in this lecture (check the topological aspect!).
So, roughly speaking, $\Spv(A)$ is an amalgamation of Riemann-Zariski spaces at all points of ${\rm{Spec}}(A)$ (but
its topology is much richer than the one on each fiber separately; ``horizontal specialization'' relative to
$\varphi$ will be a topic in a couple of lectures, 
closely related to  Theorem  \ref{thm:valgp} that is proved in the Appendix).

\medskip

\begin{eg} \label{eg:specializations} Let us fix ideas on the case $A=K$ a field. Then $\Spv(K)$, as just remarked, is nothing but $\RZ(K)$. Consider $v,w\in\Spv(A)$ with the property that $v$ is a specialization of $w$; i.e., $v\in\overline{\{w\}}$. This is equivalent to saying that all open subsets of $\Spv(A)$ containing $v$ also contain $w$, and this is in turn equivalent to the same statement, but involving just 
(finite intersections of) the basic open sets containing $v$. More explicitly, 
the condition is exactly that if a basic open set
$U\left(\frac{f}{g}\right)$ contains $v$ then it contains $w$.  The containment of $v$ says 
$$v(f)\leq v(g)\neq 0,$$
and this is meant to imply $w(f)\leq w(g)\neq 0$. Since $A=K$, this implication holds true for all $f, g \in K$ 
precisely when for any $x\in K$ we have that $v(x)\leq 1$ implies $w(x)\leq 1$. 
But this means exactly that we have a local inclusion of valuation rings:
$$R_v\subseteq R_w.$$
(Recall from Theorem \ref{thm:local} that in such cases $R_w$ is the localization of $R_v$ at a prime ideal, exactly
as in the picture for ``generization'' in algebraic geometry.) 
For example, the trivial valuation $v_0$ corresponds to the valuation ring $R_{v_0} = K$
and this contains all others, so $v_0$ is the unique generic point of $\Spv(K)$. 
\end{eg}

For $v, w \in \Spv(A)$, we say that $v$ is a {\em vertical specialization} of $w$ if it is a specialization and has the same support.
Letting $\mathfrak{p}$ denote this common support, this is exactly the specialization relation inside
the topological fiber $\varphi^{-1}(\mathfrak{p}) = \RZ(\kappa(\mathfrak{p}))$ for which
Example \ref{eg:specializations} describes the meaning of specialization in algebraic terms.

Next time we discuss quasi-compactness for $\Spv(A)$, and to do so
 it will be convenient to introduce a special class of topological spaces, as in the following definition:
\begin{defin} \label{def: spectral} A topological space $X$ is {\em spectral} if:
\begin{enumerate}
\renewcommand{\labelenumi}{\textup{(\arabic{enumi})}}
\item $X$ is quasi-compact.
\item $X$ is sober.
\item There exists a basis of quasi-compact open subsets of $X$ stable under finite intersections (which implies that
any finite intersection of  quasi-compact open subsets is quasi-compact; i.e., $X$ is a ``quasi-separated'' topological space).
\end{enumerate}
\end{defin}

The above definition is motivated by well-known topological properties of the spectrum of a ring. 


\subsection{Appendix: value groups of composite valuation rings}

We now provide a proof of Theorem \ref{thm:valgp}, using notation from that result.
We call $\overline{v}'$ the valuation on $k=R/\fm$ with valuation ring $\overline{R}'$.

Recall that the value group for $v$ is $K^{\times}/R^{\times}$ with the usual ordering that expresses divisibility relative to $R$, and the value group for $\overline{v}'$ is $k^{\times}/(\overline{R}')^{\times}$ with order structure expressing divisibility relative to $\overline{R}'$.  Finally, the value group for $R'$ is $K^{\times}/(R')^{\times}$ with order structure expressing divisibility relative to $R'$.  Note that since $R'$ is contained in $R$, $v'$ is a specialization of $v$ in $\Spv(K) = \RZ(K)$, as explained in Example \ref{eg:specializations}.

There is an evident exact sequence of abelian groups
$$1 \to R^{\times}/(R')^{\times} \to K^{\times}/{R'}^{\times} \to K^{\times}/R^{\times} \to 1$$
and under the quotient map $R \to k$ we have $R^{\times} \to k^{\times}$ is surjective by locality of $R$ and also $R' \to \overline{R}'$ is a surjective local homomorphism, so also ${R'}^{\times} \to{\overline{R}'}^{\times}$ is. Thus, 
$$R^{\times}/{R'}^{\times} \to k^{\times}/{\overline{R}'}^{\times}$$
is an isomorphism.  Hence, we have a short exact sequence of groups
$$(*)\ \ \  1 \to k^{\times}/{\overline{R}'}^{\times} \to K^{\times}/{R'}^{\times} \to K^{\times}/R^{\times} \to 1$$
expressing the value group for the composite valuation $v'$ as an extension of the value group for $v$ by the one for $\overline{v}'$.\\

But what about the relationship among the {\em order structures} for the three groups in $(*)$?
The subgroup $k^{\times}/{\overline{R}'}^{\times}$ of the totally ordered abelian group 
$K^{\times}/(R')^{\times}$ is not any old subgroup: we will show that it satisfies the following condition.

\begin{defin} Let $\Gamma$ be a totally ordered abelian group, and $H$ a subgroup.  We say $H$ is {\em convex} if for any $h, h'$ in $H$ every $\gamma$ in $\Gamma$ satisfying $h \leq \gamma \leq h'$, then necessarily $\gamma \in H$.  (That is, an element of $\Gamma$ sandwiched between two elements of $H$ is also in $H$).
\end{defin}

The interesting thing with convexity is that it is exactly what we need to have a well-defined structure of totally ordered abelian group on the quotient $\overline{\Gamma} := \Gamma/H$.  Namely, we want to define $\overline{\Gamma}_{\leq 1}$ to be the image of $\Gamma_{\leq 1}/H$, which is to say that for elements $g$ and $g'$ in $\overline{\Gamma}$ we want to say $g \leq g'$ precisely when 
$g/g' \in \Gamma_{\le 1} H$.   

It is clear that this is a translation-invariant relation that is transitive and reflexive, and certainly for any $g, g' \in \overline{\Gamma}/H$ we have $g \leq g'$ or $g' \leq g$ (since the same holds at the level of choices of representatives in $\Gamma$).  The key point is to check that this is anti-symmetric (i.e., $g \leq g'$ and $g' \leq g$  imply $g = g'$), and that holds precisely because $H$ is convex:  the task is to show that 
$$\Gamma_{\leq 1}H \cap \Gamma_{\geq 1}H = H,$$
which is to say that if $\gamma, \gamma' \in \Gamma_{\leq 1}$ and $\gamma/\gamma' = h \in H$ then $\gamma, \gamma' \in H$.  
We have  $1 \le \gamma = \gamma' h \le h$, so by convexity $\gamma \in H$, so $\gamma' = \gamma/h \in H$ too.  

In fact, convexity is also 
{\em necessary} for this to be a total order (really to be anti-symmetric), as follows.  Suppose $h \leq \gamma \leq h'$ with $h, h' \in H$ and $\gamma \in \Gamma$, so $h/\gamma \in \Gamma_{\leq 1}H$ and $\gamma/h' \in \Gamma_{\leq 1}H$. Thus, the class of $\gamma$ in the ``residue order'' is both $\geq 1$ and $\leq 1$.  Hence, if anti-symmetric would hold then this class must be $1$, which is to say $\gamma \in H$.  That is exactly the property of convexity.

Coming back to $(*)$, we cannot expect to express the ordered value group of the composite valuation $v'$ in terms of those of $v$ and 
$\overline{v}'$, as $(*)$ doesn't generally split even as abelian groups. 
Instead, we claim that the left term in $(*)$ is carried order-isomorphically
onto a convex subgroup of the middle term and that the order structure on the right term is thereby induced from the middle via the general definition given above for quotients by convex subgroups.  

As a first step, we check that the inclusion
$$k^{\times}/{\overline{R}'}^{\times} = R^{\times}/{R'}^{\times} \hookrightarrow K^{\times}/{R'}^{\times}$$
is order-preserving and in fact an order-isomorphism onto its image.
This amounts to checking that for elements $r_1, r_2 \in R^{\times}$ with reductions $\overline{r}_1, \overline{r}_2 \in k^{\times}$,
we have $\overline{r}_1/\overline{r}_2 \in \overline{R}'$ if and only if $r_1/r_2 \in R'$.  But this is obvious since
$R^{\times} \rightarrow k^{\times}$ is a homomorphism and 
$R'$ is defined to be the preimage of $\overline{R}'$ under $R \twoheadrightarrow k$.

In the order structure on the quotient term $K^{\times}/R^{\times}$ in $(*)$, the part $\leq 1$
 is exactly $(R - \{0\})/R^{\times}$, and for 
$H := k^{\times}/{\overline{R}'}^{\times}= R^{\times}/{R'}^{\times}$ we have 
$$(K^{\times}/{R'}^{\times})_{\leq 1} H = ((R' - \{0\})/{R'}^{\times})H = ((R' - \{0\})R^{\times})/{R'}^{\times}= (R - \{0\})/{R'}^{\times},$$
the final equality expressing that the inclusion $(R' - {0})R^{\times} \subset R - \{0\}$ is an equality since $\fm_R - \{0\} \subset R -\{0\}$ whereas $R - \fm_R = R^{\times}$.  Hence, by the necessity established above it follows that the image of
$H = \Gamma_{\overline{R}'}$
in $\Gamma_{R'} = K^{\times}/{R'}^{\times}$ has to be convex (as one could also check by bare hands) and we've just shown the quotient total ordering induced on $\Gamma_R$ is the one imposed on the value group of 
$v$ from that of the composite valuation $v'$ via the value group of $\overline{v}'$.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\small



\end{document}

